Author’s response to reviews

Title: Incidence and Risk Factors of Retinopathy of Prematurity in Iran: A Systematic Review and Meta-Analysis

Authors:
Gholamreza Badfar (gh_badfar@yahoo.com)
Zahra Jaafari (zahra.jaafari24@gmail.com)
Shoboo Rahmati (shoboorahmati2014@gmail.com)
Afsar Dastjani Farahani (Afsardf@yahoo.com)
Milad Azami (Milad98azami@gmail.com)

Version: 2 Date: 20 Dec 2017

Author’s response to reviews:

Dear Editor, BMC Ophthalmology

We would like to thank you for the letter, and the opportunity to resubmit a revised copy of this manuscript. We would also like to take this opportunity to express our thanks to the reviewers for the positive feedback and helpful comments for correction and modification.

We believe have resulted in an improved revised manuscript, which you will find uploaded alongside this document. The manuscript has been revised to address the reviewer comments, which are appended alongside our responses to this letter.

As we know retinopathy of prematurity (ROP) refers to the developmental disorder of the retina in premature infants and is one of the most serious and most dangerous complications in premature infants. The incidence of ROP in Iran is different in various parts of Iran and its prevalence is reported to be 1–70% in different regions. This study aims to determine the incidence and risk factors of ROP in Iran. I want you to do us a favor and consider kindly our revised manuscript for publication in your very great academic journal.

All authors have read the revised manuscript and they are in agreement with its contents. Also, all authors declare no conflict of interests.

Answers to every single question sent to the attached file.

Dr. Milad Azami
On behalf of the authors
School of Medicine, Ilam University of Medical Sciences.
Ilam, Iran.